Author's response to reviews

Title: Screening and brief interventions for hazardous and harmful alcohol use among patients with active tuberculosis attending primary care clinics in South Africa: a cluster randomized controlled trial protocol

Authors:

Karl K Peltzer (kpeltzer@hsrc.ac.za)
Pamela P Naidoo (pnaidoo@hsrc.ac.za)
Gladys G Matseke (gmatseke@hsrc.ac.za)
Khangelani K Zuma (kzuma@hsrc.ac.za)

Version: 2 Date: 20 May 2011

Author's response to reviews:

Pan-African Trial Register Number added: PACTR201105000297151